Author's response to reviews

Title: Risk Factors for Methamphetamine Use in Youth: A Systematic Review

Authors:

Kelly Russell (krussell@ualberta.ca)
Donna M Dryden (ddryden@ualberta.ca)
Yuanyuan Liang (yliang@ualberta.ca)
Carol Friesen (carol_friesen@hotmail.com)
Kathleen O'Gorman (kogorman@ualberta.ca)
Tamara Durec (Tamara.Durec@capitalhealth.ca)
T Cameron Wild (cam.wild@ualberta.ca)
Terry P Klassen (terry.klassen@ualberta.ca)

Version: 2 Date: 20 May 2008

Author's response to reviews: see over
May 20, 2008

Dear BMC Pediatrics,

Enclosed is a manuscript entitled “**Risk Factors for Methamphetamine Use in Youth: A Systematic Review**”. This paper is only being submitted to *BMC Pediatrics* and will not be submitted elsewhere while being considered by *BMC Pediatrics*. This paper has not been published elsewhere, and should it be published in *BMC Pediatrics*, it will not be published elsewhere without the editors’ permission.

All authors take full responsibility for the reported research. The authors have participated in the concept, design, analysis, interpretation, and drafting or revising of the manuscript. Each author’s contribution has been outlined in the manuscript.

To the best of our knowledge, there has yet to be a systematic review examining the risk factors for methamphetamine use among children. This systematic review employed rigours methodology and provides a concise and thorough review of toilet training methods. In addition, we are not aware of any financial or other relationships that might lead to a conflict of interest.

As requested, we have added a ‘Competing Interests and Authors Contributions section. We have also submitted the full literature searches, including the resources searched and their associated literature search terms.

Thank you in advance for considering the executive summary. We look forward to hearing from you.

Best regards,

Kelly Russell, MSc

Donna M Dryden, PhD

Yuanyuan Liang, PhD

Carol Friesen, MA, MLIS

Kathleen O’Gorman, BSc

Tamara Durec, BSc(Pharm), MLIS

T. Cameron Wild, PhD

Terry P Klassen, MD, MSc